Author's response to reviews

Title: Procalcitonin-guided antibiotic use versus a standard approach for acute respiratory tract infections in primary care: study protocol for a randomised controlled trial and baseline characteristics of participating general practitioners [ISRCTN73182671]

Authors:

Matthias Briel (brielm@uhbs.ch)
Mirjam Christ-Crain (mjchrist@uhbs.ch)
Jim Young (youngj@uhbs.ch)
Philipp Schuetz (schuetzp@uhbs.ch)
Peter Huber (huberp@uhbs.ch)
Pierre Periat (pperiat@solnet.ch)
Heiner C Bucher (hbucher@uhbs.ch)
Beat Muller (bmueller@uhbs.ch)

Version: 3 Date: 15 August 2005

Author's response to reviews: see over
Procalcitonin-guided antibiotic use versus a standard approach for acute respiratory tract infections in primary care: study protocol for a randomised controlled trial and baseline characteristics of participating general practitioners [ISRCTN73182671]

Dear Editor

We would like to thank you for reviewing our manuscript and accepting it for publication in BMC Family Practice. We carefully checked the manuscript for typing errors and made all the requested formatting changes. All authors approved of the final version.

In case of any further requests regarding the format of the manuscript we are gladly at your service.

Yours sincerely

Dr. Matthias Briel
Basel Institute for Clinical Epidemiology
University Hospital Basel
CH-4031 Basel, Switzerland

Phone 0041-61-265 3100
Fax 0041-61-265 3109
E-mail mbriel@uhbs.ch